\begin{table}[h]
\centering
\resizebox{\linewidth}{!}{%
\begin{tabular}{lcllcll}
\toprule
Benchmark & Size & Languages & Evaluation Type & M & Source & Difficulty \\
\midrule

GSM8k \cite{cobbe2021gsm8k} & 8,500 & EN & Numeric Answer & $\times$ & Crowdsourced problems & \diffbox{green!60}{Grade School} \\

MATH \cite{hendrycks2021math} & 12,500 & EN & Numeric Answer & $\times$ & Competitions / textbooks & \diffbox{yellow!70!orange}{High School} \\

MATH-Vision \cite{wang2024mathvision} & 3,040 & EN & Expression / Proof & $\checkmark$ & Math Competitions & \diffbox{orange!80}{High School} \\

CMMLU \cite{li2024cmmlu} & {11,528} & ZH & MCQ & $\times$ & Chinese exam materials & \diffbox{yellow!70!orange}{High School / College} \\

MMLU \cite{hendrycks2021mmlu} & {15,908} & EN & MCQ & $\times$ & College / professional exams & \diffbox{orange!80}{College-Level} \\

C-Eval \cite{huang2023ceval} & {13,948} & ZH & MCQ & $\times$ & Chinese college exams & \diffbox{orange!80}{College Entrance} \\

MMMU \cite{yue2024mmmu} & {11,500} & EN & MCQ / Expression & $\checkmark$ & Multimodal academic exams & \diffbox{orange!80}{College-Level} \\

AGIEval \cite{zhong2024agieval} & 3,300 & EN \& ZH & MCQ / Expression & $\times$ & College entrance exams & \diffbox{orange!80}{College Entrance} \\

JEEBench \cite{arora2023jeebench} & 515 & EN & MCQ / Numeric Answer & $\times$ & Indian JEE Advanced & \diffbox{orange!90}{JEE Advanced Exam} \\
\midrule
OlympiadBench \cite{he2024olympiadbench} & 6,142 & EN \& ZH & Proof / Expression & $\checkmark$ & Official Websites & \diffbox{red!70}{Olympiad Level} \\

OlympicArena \cite{huang2024olympicarena} & 3,233 & EN \& ZH & Proof / Process & $\checkmark$ & Official Websites & \diffbox{red!70}{Olympiad Level} \\

Omni-Math \cite{gao2024omnimath} & 4,428 & EN & Proof / Process & $\times$ & AoPS Forum / Contest Pages & \diffbox{red!70}{Olympiad Level} \\


IneqMath \cite{sheng2025solving} & 1,552 & EN & Proof / Analytical Tools & $\times$ & Curated Inequalities Problems & \diffbox{red!70}{Olympiad Level} \\

OlymMATH \cite{sun2025challenging} & 200 & EN \& ZH & Numeric Answer  & $\times$ & AoPS Forum/Official Websites & \diffbox{red!70}{Olympiad Level} \\

LiveAoPS \cite{mahdavi2025leveraging} & \textit{-} & EN & Numeric / Expression& $\times$ & AoPS Forum (rolling snapshot) & \diffbox{red!70}{Olympiad Level} \\

MathArena \cite{balunovic2025matharena} & 162 & EN & Final Answer / Proof & $\checkmark$ & Newly released competitions & \diffbox{red!70}{Olympiad Level} \\

IMOBench \cite{luong2025towards} & 460 & EN & Numeric / Proof  & $\times$ & IMO \& national archives & \diffbox{red!70}{Olympiad Level} \\

\midrule

{\textbf{MathNet} (ours)} & & EN, ZH, ES &  &  & Official Country  Booklets/  &  \\
 & {\textbf{30,767}} & RU, FR, RO  & Expression / Proof &  $\checkmark$ &International and National & \diffbox{red!80}{Olympiad Level} \\

 &  & and 11 other &  &  & Contests &  \\
\bottomrule
\end{tabular}%
}
\caption{Comparison of mathematical reasoning benchmarks across different sizes, languages, evaluation types, and difficulty levels. 
We include both unimodal and multimodal datasets, spanning grade-school to Olympiad-level mathematics. Our Proposed \textsc{MathNet} expands coverage to 17 languages and focuses on proof- and process-based evaluation.% with authentic national contest problems.
}
\label{tab:math-benchmarks}
\end{table}